\documentstyle[11pt,simplemargins]{article}
\setallmargins{1in}
% no page numbers
\pagestyle{empty}
% counter for page numbers
\newcounter{pg}
\setcounter{pg}{2}

% DEFINE THE PROBLEM SET NUMBER
\def\psnum{3, part 1}

% DEFINE THE DUE DATE
\def\due{March 18, 1992}

% title command
\newcommand{\mytitle}{{\large\noindent 6.863J Natural Language
Processing \hfill{} Juliet Bourne \\ 
Lab \#\psnum \hfill{} \due \\}
\vspace{-.43in}
\begin{center} \rule{6.5in}{.25mm} \end{center}}
% newpage setup command 
\newcommand{\mypage}[1]{\newpage{\large\noindent 6.863J Natural Language
Processing \hfill{} Juliet Bourne \\
Lab \#\psnum \hfill{} \due \\}
\vspace{-.43in}
\begin{center} \rule{6.5in}{.25mm} \end{center}}
% problem command
\newcommand{\prob}[1]{\vspace*{.15in}\noindent{\fbox{\large\bf #1}}}
% subproblem command
\newcommand{\subprob}[1]{\vspace{.15in} \noindent{\large\bf #1.}}
% equations commands
\newcommand{\beq}{\vspace*{-.3in}\begin{eqnarray*}}
\newcommand{\eeq}{\end{eqnarray*}}

\begin{document}

\mytitle

\noindent Miniature digram probability table covering eight lexical
categories: 

\vspace*{.15in}
\begin{center}
\begin{tabular}{|l||cccccccc|}\hline
\mbox{} & det & prep & adj & cnoun & pron & poss & RB & sup \\ \hline\hline
det & 1,A & 1,A & 3,C & 4,C & 1,A & 1,A & 1,A & 2,C \\
prep & 4,C & 3,B & 3,C & 3,C & 3,C & 3,C & 2,B & 2,B \\
adj & 2,B & 2,B & 2,C & 4,C & 2,B & 2,D & 2,B & 1,A \\
cnoun & 2,E & 3,C & 2,B & 3,C & 2,C & 2,B & 2,B & 2,E \\
pron & 3,B & 2,B & 2,B & 3,B & 2,B & 2,B & 2,B & 2,D \\
poss & 1,A & 1,A & 3,C & 4,C & 1,A & 2,B & 1,A & 2,C \\
RB & 3,B & 2,B & 2,B & 3,B & 2,B & 3,B & 2,B & 2,B \\
sup & 2,B & 3,C & 3,C & 4,C & 2,B & 2,E & 2,B & 2,B \\ \hline
\end{tabular}
\end{center}

\vspace*{.25in}
\noindent Examples:

\begin{itemize}
\item prep-prep : ``I went {\bf down to} New York this weekend.''
\item prep-RB : ``He fell {\bf down quickly}.''
\item prep-sup : ``They showed us the next {\bf to smallest} house.''
\item adj-det : ``I was {\bf happy the} ring was found.''
\item adj-prep : ``I was {\bf happy for} the winners.''
\item adj-pron : ``I was {\bf happy I} didn't fall.''
\item adj-poss : ``It was {\bf hot, my} foot!''
\item adj-RB : ``She turned {\bf red quickly}.''
\item cnoun-det : ``{\bf Money, the} only thing he ever loved.''
\item cnoun-adj : ``They painted the {\bf town red}.''
\item cnoun-pron : ``We displayed the {\bf sculpture we} made.''
\item cnoun-poss : ``The {\bf dog, my} foot!''
\item cnoun-RB : ``He took out the {\bf gun slowly}.''
\item cnoun-sup : ``That {\bf princess, prettiest} of them all, got
away.''
\item pron-det : ``She gave {\bf me the} papers to read.''
\item pron-prep : ``They took {\bf him to} his new office.''
\item pron-adj : ``I gave {\bf him blue} pens to play with.''
\item pron-cnoun : ``He showed {\bf me pictures} he drew.''
\end{itemize}

\mypage

\begin{itemize}
\item pron-pron : ``It was {\bf me he} didn't like.''
\item pron-poss : ``He gave {\bf me his} notebook.''
\item pron-RB : ``He waited for {\bf them quietly}.''
\item pron-sup : ``And then {\bf they, lowest} of the low, walked out.''
\item poss-poss : ``{\bf My, his}, and their plans were all different.''
\item RB-det : ``He saw {\bf quickly the} error of his ways.''
\item RB-prep : ``She ran {\bf quickly to} her class.''
\item RB-adj : ``The leaves turned {\bf slowly brown}.''
\item RB-cnoun : ``They sang {\bf loudly songs} of merriment.''
\item RB-pron : ``{\bf Stupidly, they} gave him the password.''
\item RB-poss : ``Tell {\bf me quickly} my name.''
\item RB-RB : ``The cat stole {\bf quickly, silently} into the room.''
\item RB-sup : ``He ran {\bf steadily, last} in his group.''
\item sup-det : ``The {\bf largest, the} most determined player yet.''
\item sup-pron : ``Being the {\bf youngest, I} went last.''
\item sup-poss : ``The {\bf largest, my} foot!''
\item sup-RB : ``He became the {\bf quickest slowly}.''
\item sup-sup : ``The {\bf biggest, meanest} IRS agent knocked on my door.''
\end{itemize}

\end{document}

